Author’s response to reviews

Title: A Huge Benign Fibrous Histiocytoma Arising from the Renal Capsule: Report of a Case

Authors:

sun chao (414411899@qq.com)
wang shengli (victory1030@sina.com)
li bo (16693915@qq.com)
sun xueting (sxt8443@sina.com)

Version: 1 Date: 11 Jan 2019

Author’s response to reviews:

1. This article failed to make its novelty or scarcity of this case clear. The value of this article should be more expressed.

   Respondence: We have already emphasized the novelty and scarcity of this case. These can be found in Abstract background section, line 9, page 1; Abstract conclusions section, line 19, page 1; Discussion and Conclusions section, line 12-13, page 6.

2. In order to increase the value of this article, it is recommended to add a table which summarizes previous case reports and this case.

   Respondence: Our intention is to report a case, not to write a summary or review. At one time we thought of adding a table which summarizes previous case reports, but later we considered it would make the article too long, so we gave up.